---
sidebar_position: 1
slug: /
title: Jonas_Jones Documentation
---

import Translate from '@site/src/components/Translate';

# <Translate id="pages.content.index.title" />

:::info
Here is all the documentation to my *personal* projects.

*All projecets managed by Organizations have their separate docs.*
:::

## <Translate id="pages.content.index.subtitle" />
Check out the sidebar or here, below:
<div className="page-content-flex-container">
  <div className="page-content-flex-box page-content-flex-box1">
  ### [Jonas_Jones API](/api)
  The API endpoint used for many of my projects
  </div>
  <div className="page-content-flex-box page-content-flex-box2">
  ### [My Minecraft Mods](/overview/mods)
  An Overview of all my Minecraft Mods
  </div>
</div>
<div className="page-content-flex-container">
  <div className="page-content-flex-box page-content-flex-box1">
  ### [TurboOctoPotato](/turbo-octo-potato)
  Guides to all my random Scripts
  </div>
  <div className="page-content-flex-box page-content-flex-box2">
  ### [Achievement Guides](/overview/achievement-guides)
  An Overview of all my Achievement Guides
  </div>
</div>
<div className="page-content-flex-container">
  <div className="page-content-flex-box page-content-flex-box1">
  ### [3D Printables](/overview/3d-printables)
  Guides and Showcases of all my 3D Printables
  </div>
  <div className="page-content-flex-box page-content-flex-box2">
  ### [Services & Platforms](/overview/services-and-platforms)
  An Overview of all my Services and Platforms
  </div>
</div>
